Author’s response to reviews

Title: Changes in Anti-Viral Effectiveness of Interferon After Dose Reduction in Chronic Hepatitis C Patients: a Case Control Study

Authors:

Frank C Bekkering (fcbekkering@hotmail.com)
Avidan U Neumann (neumann@mail.biu.ac.il)
Johannes T Brouwer (brouwer@mdl.azr.nl)
Rachel S Levi-Drummer (drummer@mail.biu.ac.il)
Solko W Schalm (schalm@mdl.azr.nl)

Version: 2 Date: 4 Dec 2001

On page 11, "HCVRNA" has been replaced with "HCV RNA". On the same page, "virussuppressive" has been replaced with "virus suppressive".

Both changes are according to suggestions made by the second reviewer.